Author's response to reviews

Title: The use of health-related quality of life (HRQOL) in children and adolescents as an outcome criterion to evaluate family oriented support for young carers in Germany: an integrative review of the literature

Authors:

Jörg große Schlarmann (schlarmann@uni-wh.de)
Sabine Metzing-Blau (smetzing@uni-wh.de)
Wilfried Schnepp (schnepp@uni-wh.de)

Version: 6 Date: 11 December 2008

Author's response to reviews:

We would like to thank the reviewers for their feedback on our manuscript.

Referee 3: As recommended, we removed the last sentence of the abstract.

Referee 2:
We agree that it might be confusing to have both negatively as well as positively connotated phrases. But since they are quotations from a assessment instrument on the one hand and study results on the other hand, we prefer to not change the paragraphs.